Author's response to reviews

Title: Attitudes towards the administration of long-acting antipsychotics: a survey of physicians and nurses

Authors:

   Paul Geerts (pgeerts@its.jnj.com)
   Guadalupe Martinez (gmartin5@its.jnj.com)
   Andreas Schreiner (aschrein@its.jnj.com)

Version: 2 Date: 27 February 2012

Author's response to reviews: see over
Dear Ms Olino

MS: 1848277921663815
Attitudes towards the administration of long-acting antipsychotics: a survey of physicians and nurses
Paul Geerts, Guadalupe Martinez and Andreas Schreiner.

Thank you for your query regarding the acknowledgements statement in the manuscript detailed above. I can confirm that Dr Marion James is aware that she has been acknowledged for her contribution to the manuscript.

The data reported in this manuscript are owned by Janssen. Research International received payment from Janssen to collect the data on their behalf. A statement to this effect has been added to the manuscript.

Please let me know if any additional information is required.

Yours sincerely,

Andreas Schreiner